Chohan /31-Oct-05
Week 44 starting 31Oct 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  24Oct 05 :  10/31/2005 10:20:08 AM
	1
	A1
	MBBL3409
	ICS 5 Traveller Last
	295.18

	9
	A2
	MBBR3340
	Prep 5 Cool Down
	171.95

	10
	B1
	SSS653
	Prep 1 Lyre Test
	294.64

	5
	B2
	SSS166
	Prep 0 Scan Bench
	295.8

	11
	C1
	MBBL3407
	Prep 2.1 IAP @ warm LF
	293.57

	6
	C2
	MBBR3415
	ICS 5 Traveller Last
	295.8

	4
	D1
	SSS272
	PT 9 CDAP HV
	1.89

	3
	D2
	SSS631
	Prep 5 Cool Down
	1.90

	7
	E1
	MBAR1270
	Prep 5 Cool Down
	21.85

	2
	E2
	MBAL1072
	PT 12 Warm Up
	272.35

	12
	F1
	MBAR1264
	ICS 4 Final connection
	294.1

	8
	F2
	MBAL3395
	PT 13.2 Resist. Meas.
	296.88


   Tempr. from TBSM18 site
 
   
Coordinator this week: Vittorio; MTM support: 16-3602

Only 15 complete Magnet Tests cycles  from 11 benches ( E2 was still under CFB repair)   10 Dipoles + 3  SSS  + 2 SSSS

 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 

Situation From Cryogenics:  J-P Lamboy or ACR Rep.

E2 CFB (leak) is now Repaired, 4 weeks down time 29-sept-05 till 27-oct-05.
Special  Points :
· Connection Problems for Sp SSS on B1 (ICS + )

· 3139 re-test and signals
· Special SSS631  on D2 :   SSW & Chaconsa ? (pgm Marco)
· SSW   problems on dipole 3409  &  MM cancelled( Marco+Stephane).
AOB 
	Magnets under Test from 23-10-2005 to 29-10-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	
	

	1
	TBA1
	MBAR1275 
	[image: image1.png]



	21-10-2005 
	33 h.
	27 h.
	23 h.
	15 h.
	17 h.
	26-10-2005 
	115 h.
	Stripping
	
	
	

	2
	TBA2
	MBAL1154 
	[image: image2.png]



	19-10-2005 
	28 h.
	44 h.
	27 h.
	16 h.
	14 h.
	24-10-2005 
	129 h.
	Stripping
	
	
	

	3
	TBA2
	MBBL3408 
	[image: image3.png]



	24-10-2005 
	25 h.
	23 h.
	44 h.
	19 h.
	7 h.
	29-10-2005 
	118 h.
	Stripping
	
	
	

	4
	TBB1
	SSS638 
	[image: image4.png]



	20-10-2005 
	31 h.
	28 h.
	77 h.
	19 h.
	25 h.
	28-10-2005 
	180 h.
	Stripping
	
	
	

	5
	TBB2
	SSS150 
	[image: image5.png]



	20-10-2005 
	29 h.
	19 h.
	19 h.
	9 h.
	13 h.
	23-10-2005 
	89 h.
	Stripping
	
	
	

	6
	TBB2
	SSS162 
	[image: image6.png]



	23-10-2005 
	33 h.
	17 h.
	22 h.
	7 h.
	10 h.
	27-10-2005 
	89 h.
	Stripping
	
	
	

	7
	TBC1
	MBAL2192(2) 
	[image: image7.png]



	21-10-2005 
	0 h.
	20 h.
	9 h.
	15 h.
	20 h.
	24-10-2005 
	64 h.
	Stripping
	
	
	

	8
	TBC2
	MBBL3405 
	[image: image8.png]



	22-10-2005 
	28 h.
	22 h.
	19 h.
	12 h.
	13 h.
	26-10-2005 
	94 h.
	Stripping
	
	
	

	9
	TBD1
	SSS292(2) 
	[image: image9.png]



	22-10-2005 
	0 h.
	21 h.
	91 h.
	8 h.
	20 h.
	28-10-2005 
	140 h.
	Stripping
	
	
	

	10
	TBD2
	SSS604 
	[image: image10.png]



	11-10-2005 
	63 h.
	23 h.
	175 h.
	72 h.
	18 h.
	25-10-2005 
	351 h.
	Stand By
	
	
	

	11
	TBE1
	MBAL3708 
	[image: image11.png]



	25-10-2005 
	27 h.
	19 h.
	21 h.
	14 h.
	12 h.
	29-10-2005 
	93 h.
	Stripping
	
	
	

	12
	TBE1
	MBBL3404 
	[image: image12.png]



	19-10-2005 
	27 h.
	47 h.
	22 h.
	15 h.
	21 h.
	25-10-2005 
	132 h.
	Stripping
	
	
	

	13
	TBF1
	MBAR2236 
	[image: image13.png]



	20-10-2005 
	24 h.
	46 h.
	5 h.
	19 h.
	24 h.
	25-10-2005 
	118 h.
	Stand By
	
	
	

	14
	TBF2
	MBAL2198 
	[image: image14.png]



	18-10-2005 
	27 h.
	25 h.
	21 h.
	15 h.
	8 h.
	23-10-2005 
	96 h.
	Stripping
	
	
	

	15
	TBF2
	MBAL2210 
	[image: image15.png]



	23-10-2005 
	31 h.
	26 h.
	16 h.
	13 h.
	19 h.
	27-10-2005 
	105 h.
	Stripping
	
	
	

	
	
	
	
	
	
	
	
	
	
	
	
	
	
	
	


	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10

	1
	MBAR1275
	12204.3 A.
(D1-L-S01)
	12596.3 A.
(D1_U_S06)
	[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-

	2
	MBAL1154
	12432 A.
(D2_L)
	12696.5 A.
(D2_L)
	12726.3 A.
(D1-L)
	[image: image17.png]



	-
	-
	-
	-
	-
	-

	3
	MBBL3408
	12049.3 A.
(D2_U)
	12759.5 A.
(D1_L)
	12536.9 A.
(D1_L)
	[image: image18.png]



	-
	-
	-
	-
	-
	-

	4
	SSS638
	5017.2 A.
(MQML-E12)
	5601.3 A.
(MQML_I34)
	5820 A.
[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-

	5
	SSS150
	12545.9 A.
(QI-12)
	12639.3 A.
(QI-34)
	12800 A.
[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-

	6
	SSS162
	12514 A.
(QI-34)
	12843 A.
[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAL2192
	11907.8 A.
(D2_U)
	12773 A.
(D2 - U)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-

	8
	MBBL3405
	1234.2 A.
(D2_U)
	12274.3 A.
(D1_L)
	12792.4 A.
(D2-L)
	[image: image23.png]



	-
	-
	-
	-
	-
	-

	9
	SSS292
	11496.3 A.
(QE_34)
	12183.9 A.
(QI-12-S5)
	12527.4 A.
(QI-34-S6)
	[image: image24.png]



	-
	-
	-
	-
	-
	-

	10
	SSS604
	5409.7 A.
(MQM1-Ext)
	5820 A.
[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL3708
	12396.7 A.
(D2_L)
	12387.1 A.
(D1-U)
	12857 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-

	12
	MBBL3404
	9498.7 A.
(D2-U)
	11871.7 A.
(D2-U)
	12203.1 A.
(D2-U)
	12627.1 A.
(D2-U)
	12639.3 A.
(D2_L)
	12832.5 A.
(D2-L)
	12850 A.
[image: image27.jpg]



	12857 A.
[image: image28.jpg]



	-
	-
	-

	13
	MBAR2236
	[image: image29.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	14
	MBAL2198
	10081.3 A.
(D1_U)
	11110.4 A.
(D2_L)
	12129.6 A.
(D1_L)
	12676.4 A.
(D2_L)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	

	15
	MBAL2210
	10804.1 A.
(D1-U)
	12421.7 A.
(D1-L)
	12389.3 A.
(D1_L)
	[image: image31.png]



	-
	-
	-
	-
	-
	-
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COHABITATION Issues  on B & D clusters as usual  due to Special SSS 604 (D2, Chaconsa , etc)    &  638( on B1 )
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_25_A
	TBD1


	EF corrector test at 600 A (MQT). Power supply deliver error message” digital parameters out of limits”.
	AB/PO
	
	Richard came and tried many times. At the end the measurements were launched.

Investigation by AB/PO Friday. Inductance loop not correct.

	Wednesday_26_M
	TBB2


	Corrector powering on SSS162 could not be started as the PC Aux interlock in PLCSE is not clearing.
	AB/PO
	
	120A power supply stop button was found in the "OFF" position

	Thursday_27_A
	TBD1


	TT821 on MB3139 is not working at cold (already reported during the first run).
	AB/PO
	
	NC opened. Richard measured the probe at the level of the IFS box. One contact is open.

	Thursday_27_A
	TBD1


	IAP @warm final on SSS292: The results on aperture voltages displayed were not correct.
	AB/CO
	IAP
	OP team checked the results with lf DAQ.

	Thursday_27_M
	TBA2


	Angle measurements on MB3408: SSW3 did not work.
	MTM
	
	Nikolay advised to change with system 1. Measurements during Thursday night; wire broke 4 times. Stephane and Marco decided to restrict the number of measurements.

	Friday_28_A
	TBA1


	MB1072: From VT measurements no diode in the magnet.
	MTM
	
	NC opned. MB1072 is in the list of diode work from D.Tommasini.

It will not influence the decision stripping/stand by.

	Friday_28_A
	TBF1


	MB2209: While launching PT5.1 a message appeared on the local server” One or more channels have < 100 ms pre-trigger time”. You have to acknowledge to arm hf and lf DAQ. 
	AB/CO
	Hf,lf
	Not blocking. Configuration corrected by Richard. Problem known for all benches.

Richard asked Maryline for a correction asp.

	Friday_28_N
	TBD1


	SSS272: Permanent trigger in VclampD and VclampF. Voltage measured VclampD=10V VclampF=-6.5V
	ICS
	
	ICS is asked to check the connection

	Saturday_29_M
	TBA1


	Leak found in MB3409.
	ACR
	
	Investigation by cryo team: Leak in the flexible of the line E. Decision was to continue the test. MB3409 leak test could not be validated. It has to be reconnected and leak test redone.



	Saturday_29_M
	TBB1


	ICS cannot connect SSS653 because of a piece to be inserted in the cover line block has not the proper thickness.
	ICS
	
	B.Poulnais gave to ICS team the correct piece.

	Sunday_30_M
	TBC1


	Corrector magnets were found open during the final resistance measurement CDW2 for the MB3139.
	MTM
	
	NC opened.

Stephane’s comment : Magnet already stripped after the first run: correctors not connected.



	Sunday_30_N
	TBD1


	RMD did not work: “PC fault” displayed.
	AB/PO
	
	Richard came and reset the Digital Controler.

	Sunday_30_N
	TBB2


	SSS166: Bar code not correct.
	
	
	NC opened.


